Author's response to reviews

Title: Cytotoxicity of Selected Cameroonian Medicinal Plants and Nauclea pobeguinii Towards Multi-Factorial Drug-Resistant Cancer Cells

Authors:

Victor Kuete (kuetevictor@yahoo.fr)
Pergaud L Sandjo (plsandjo@yahoo.fr)
Armelle T Mbaveng (armkuete@yahoo.fr)
Jackson A Seukep (seukepp@yahoo.fr)
Bonaventure T Ngadjui (btgadjui@yahoo.fr)
Thomas Efferth (efferth@uni-mainz.de)

Version: 4 Date: 13 August 2015

Author's response to reviews:

Detailed responses to reviewer's Comments for MS: 2075344625159266R2

Dear Editor, Dear Tom,

Thanks for sending the reviewer's recommendations for the above manuscript. We are now resubmitting the revised version taking in account their recommendations to improve it. The revision made are highlighted in recolor color in the manuscript and also listed point-by-point below.

I hope you will found the work suitable for publication now

best regards

(Prof. Dr. Victor Kuete)

Reviewer: Mahmoud El-Readi
Reviewer's report:

major
explain the cytotoxicity results of HepG-2 cell in the discussion otherwise exclude it from the manuscript.
Answer: data related to HeG2 were deleted as we do not have any specific explanations to the obtained results.

minor
add the spectrum charts to the supplemental data
Answer: this has been provided